Author's response to reviews

Title: Illness cognition as a predictor of exercise habits and participation in cardiac prevention and rehabilitation programs after acute coronary syndrome

Authors:

Orna Reges (orna.reges@gmail.com)
Noa Vilchinsky (Noa.Vilchinsky@biu.ac.il)
Morton Leibowitz (LeibowM@clalit.org.il)
Abdulrahem Khaskia (khaskiaa@clalit.org.il)
Morris Mosseri (Morris.Mosseri@clalit.org.il)
Jeremy D Kark (jeremy1@vms.huji.ac.il)

Version: 3 Date: 6 October 2013

Author's response to reviews: see over
October 06, 2013

Prof. Oscar Franco,
BMC Public Health

RE: Resubmission of manuscript 1674492181987400

Dear Prof. Franco,

I am enclosing the particulars of the 2 minor changes requested in our MS titled: "Illness cognition as a predictor of exercise habits and participation in cardiac prevention and rehabilitation programs after acute coronary syndrome ".

1. Please state in your revised manuscript the name of ethics committee that approved the study.

   We added the name of the ethics committee on page 7:
   
   The study was authorized by the Meir Hospital Institutional Review Board.

2. Please add a title for the Competing Interest Section

   The following information added to page 14:

   **Competing interests**
   The authors declare that they have no competing interests.

We would like to thank you again for the time and thought you have invested to improve this paper. We hope we have responded satisfactorily to your requests and that you will now consider our revised manuscript suitable for publication in the *BMC Public Health*.

Sincerely,

Orna Reges